Author's response to reviews

Title: Efficacy of a whole-body vibration intervention to improve exercise tolerance and functional performance of the lower limbs of people with chronic obstructive pulmonary disease

Authors:

Trentham P Furness (trentham.furness@acu.deu.au)
Nicole J Bate (nicole.bate@southernhealth.org.au)
Liam Welsh (liam.welsh@rch.org.au)
Geraldine A Naughton (geraldine.naughton@acu.edu.au)
Christian Lorenzen (christian.lorenzen@acu.edu.au)

Version: 3 Date: 4 July 2012

Author's response to reviews: see over
To the Editorial Board of BMC Pulmonary Medicine,

Please find our revised manuscript for study protocol (MS: 5340789706872799) entitled;

Efficacy of a whole-body vibration intervention to improve exercise tolerance and functional performance of the lower limbs of people with chronic obstructive pulmonary disease

- Proofs of HREC approval were emailed to the Editorial office on 28/06/2012
- The trial was not funded in any way
- The trial registration number was added to the abstract

Thank you with your persistence with our work. The efforts of the Editorial Office are immensely appreciated.

Regards,

Mr. Trentham P. Furness (PhD Candidate)

04/07/2012